Author's response to reviews

Title: Why not just Google it? An assessment of information literacy skills in a biomedical science curriculum.

Authors:

Karl Kingsley (karl.kingsley@unlv.edu)
Gillian M Galbraith (gillian.galbraith@unlv.edu)
Matthew Herring (matthew.herring@sdm.unlv.edu)
Eva Stowers (eva.stowers@unlv.edu)
Tanis Stewart (tastewart@fdic.gov)
Karla V Kingsley (karlak@unm.edu)

Version: 7 Date: 19 April 2011

Author's response to reviews:

Four minor typographical errors were found, that included:

1. A dash "-" was removed from page 2 in the author contact details;
2. A comma "," was removed from page 8 in the Human Subjects Exemption section;
3 and 4. Journal titles were corrected for references 19 and 20;
   Thank you.